# RESEARCH



# Prevalence and correlates of induced abortion: results of a facility-based cross-sectional survey of parturient women living with HIV in South Africa



Anthony Idowu Ajayi<sup>1</sup>, Bright Opoku Ahinkorah<sup>2</sup>, Abdul-Aziz Seidu<sup>3,4\*</sup> and Oladele Vincent Adeniyi<sup>5</sup>

# Abstract

**Background:** There is a paucity of studies examining the prevalence and correlates of induced abortion among women living with HIV. Our study fills this gap by examining the prevalence and correlates of induced abortion among parturient women living with HIV in Eastern Cape, South Africa.

**Methods:** We analysed cross-sectional survey data of the East London Prospective Cohort Study, which took place between September 2015 and May 2016 in three large maternity facilities in the Buffalo/Amathole districts of the Eastern Cape Province of South Africa. A total of 1709 parturient women living with HIV who gave birth over the study period were recruited. We carried out descriptive and inferential statistics.

**Results:** The prevalence of induced abortion was 19% (95% CI: 17.2–20.9%), but varied by women's socio-demographic characteristics. Induced abortion prevalence was higher among women aged 25 years and over (21.4%), than among women aged less than 25 years (11.0%). Those ever married or cohabiting (26.8%) reported a higher level of induced abortion than those never-married women (15.6%). Those already diagnosed HIV positive before their index pregnancy (20.2%) had a higher prevalence of induced abortion than those diagnosed during their index pregnancy (14.1%). In the adjusted logistic regression, women were more likely to have ever induced abortion if they were ever married or cohabiting (aOR; 1.86 95% CI; 1.43–2.41), ever smoked (aOR: 1.51; 95% CI: 1.01–2.28) and diagnosed with HIV before their index pregnancy (aOR:1.44; 95% CI: 1.02–2.05) but less likely if younger than 25 years (aOR; 0.51 95% CI:0.35–0.73).

**Conclusion:** About one in five women living with HIV had ever induced abortion in the study settings, indicating that abortion service is one of the main reproductive health services needed by women living with HIV in South Africa. This is an indication that the need for abortion is somewhat high in this group of women. The finding, therefore, highlights the need for concerted efforts from all stakeholders to address the unmet need for contraception among women living with HIV to prevent unintended pregnancy.

**Keywords:** Induced abortion, Pregnancy termination, Unwanted pregnancy, Unintended pregnancy, Contraception, Women living with HIV, South Africa

\*Correspondence: abdul-aziz.seidu@stu.ucc.edu.gh

<sup>3</sup> Centre for Gender and Advocacy, Takoradi Technical University, Takoradi, Ghana

Full list of author information is available at the end of the article



© The Author(s) 2022. **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

## **Plain language summary**

There is a paucity of studies examining the prevalence and correlates of induced abortion among women living with HIV. Our study fills this gap by examining the prevalence and correlates of induced abortion among parturient women living with HIV in Eastern Cape, South Africa. We analysed a cross-sectional survey data of the East London Prospective Cohort Study, which took place between September 2015 and May 2016 in three large maternity facilities in the Buffalo/Amathole districts of the Eastern Cape Province of South Africa. A total of 1709 parturient women living with HIV who gave birth over the study period were recruited. We carried out descriptive and inferential statistics. The prevalence of induced abortion was 19%, but varied by women's sociodemographic characteristics. Induced abortion prevalence was higher among women aged 25 years and over (21.4%), ever married (26.8%), those already diagnosed HIV positive before their index pregnancy (20.2%). About one in five women living with HIV had ever induced abortion in the study settings, indicating that abortion service is one of the main reproductive health services needed by women living with HIV in South Africa. This is an indication that the need for abortion is somewhat high in this women. The finding, therefore, highlights the need for concerted efforts from all stakeholders to address the unmet need for contraception among women living with HIV to prevent unintended pregnancy.

# Introduction

According to the 2020 UNAIDS report on the global AIDS epidemic [1], women constitute 53% of all adults living with HIV. These women face several reproductive health challenges, including unmet need for contraception and unintended pregnancies [2, 3]. In sub-Saharan Africa, where HIV is endemic, 63% of women have an unmet need for sexual and reproductive health services [4]. Other studies have also found low desire for children among women living with HIV due to fear of giving birth to HIV-positive children [5–7]. According to UNAIDS estimates, unintended pregnancy rates among women living with HIV range from 51 to 90% globally [8]. Furthermore, studies from sub-Saharan Africa show that the rates of unintended pregnancies among women living with HIV range from 35 to 71% [9–13]. In South Africa, more than two-thirds of pregnancies among women living with HIV are unintended [14–16], with a higher prevalence among parturient women diagnosed with HIV during pregnancy [16]. Studies comparing the prevalence of unintended pregnancy between women living with or without HIV appear to suggest that those with HIV have a higher prevalence of unintended pregnancy [10, 13].

High unmet need for contraception and contraceptive failure are critical risk factors for unintended pregnancy among women living with HIV. Studies in South Africa have found contraceptive failure due, incorrect and inconsistent condom use, and lack of knowledge of emergency contraception as reasons for the high prevalence of unintended pregnancies among women living with HIV in the country [17, 18]. In a study conducted in Zimbabwe, 48% of women with intended pregnancies were using contraceptives at the time of their conception [13]. Another study shows that four in five women reporting contraceptive failure were using condoms, 13% oral contraceptives, and 5% injectable contraceptives [12].

The high prevalence of unintended pregnancies among women living with HIV necessitate improved access to comprehensive abortion services if requested by women. Comprehensive abortion services offer immense benefits to pregnant women, including saving lives that could have been lost due to unsafe abortions and prevention of temporary or permanent disabilities [19]. While nonuse of contraceptives or contraceptive failure may result in unwanted or mistimed pregnancies, pregnant women living with HIV may want induced abortion because of fear of transmitting the virus to the baby [20, 21], fear of exposing the child to later stigma and discrimination, fear of leaving a child orphaned [20, 21] and fear of bringing financial burden on her in relation to the care of herself and the child [22]. Since unwanted pregnancy is the main reason women seek abortion services, it is expected that induced abortionwill be common among women living with HIV, known to have a high prevalence of unintended pregnancy [15, 16].

However, studies examining the prevalence of induced abortion among women living with HIV are scarce. The few existing studies are mainly from the global north, indicating a clear gap in sub-Saharan Africa literature on abortion among women living with HIV. Existing studies, however, appear to suggest that the prevalence of induced abortion is higher among women living with HIV compared to HIV-negative women [10, 23]. For instance, Pinho et al. [24] found that the prevalence of induced abortion was 11.9% among women living with HIV/AIDS and 3.0% among women not living with HIV/ AIDS. Another study also found that the prevalence of induced abortion among women living with HIV/ AIDS was 6.5%, while women not living with HIV/AIDS was 2.9% [23]. Older age, higher education level, multiple sexual partnerships, having had children prior to the index pregnancy, and living with a sexual partner during pregnancy were identified as factors associated with the high prevalence of induced abortion among women living with HIV [23, 24].

Although the choice on termination on pregnancy Act 1996 gives South African women, regardless of age or marital status, the right to access comprehensive abortion services within the first 12 weeks of pregnancy, there are still challenges women face in procuring abortion, such as social stigma, thus, making women patronize unsafe backstreet induced abortions [25, 26]. Notwithstanding the restrictive socio-cultural environment in South Africa, women have the safe option of terminating a pregnancy. Research among pregnant women living with HIV who sought abortions has indicated problems they encounter when accessing safe abortion services, including faulty information and discriminatory treatment from healthcare providers [20, 21, 27]. Studies in South Africa have cited delays in healthcare provision and stigma as significant factors that hinder women living with HIV from seeking safe abortion care services [21, 22]. A few women were misinformed that only one legal abortion is allowed per woman [21]. A report by the South African National AIDS Council cited sterilization as a requirement from HIV-positive parturient women for obtaining a legal pregnancy termination [28]

To the best of our knowledge, there is not study that has estimated the prevalence and correlates of induced abortion among women living with HIV in South Africa. Our study fills this gap by examining the prevalence and correlates of induced abortion among parturient women living with HIV in South Africa. The findings of our study could inform health policies and planning regarding the provision of abortion care in South Africa (Tables 1 and 2).

#### Methods

# Study design and setting

We analysed the baseline data of the East London Prospective Cohort Study, a cross-sectional study, which was conducted between September 2015 and May 2016 in three major maternity facilities in the Buffalo/Amathole districts of the Eastern Cape Province of South Africa [9, 29]. These health facilities serve a total population of 1,674,637 residents, with the Amathole district (892,637) marginally outnumbering the Buffalo City metropolitan area (755,000) [32]. The prevalence of HIV was 12.7% in the general population and 30% among pregnant women in the study setting [32]. The three health facilities chosen reflect the province's diverse demographics and levels of health care. For example, Frere hospital is an urban tertiary health facility that receives patients from all over the province, while Cecilia Makiwane hospital, a regional health facility, is situated in the semi-urban Mdantsane township and offers both levels one and two services in the region. Bisho Hospital is a district health facility that mainly serves rural communities in Bisho and the surrounding areas. The study's detailed methodology has already been published in previous studies [9, 16].

#### Participants and sample size

The East London Prospective Cohort study was established to track PMTCT outcomes and associated behavioural factors with a view to improving quality of care in the study setting. It is made up of women diagnosed with HIV receiving maternity care in three health facilities. The sample size for the East London Prospective Cohort study was estimated using the approximate proportion of parturient women living with HIV on treatment one year after delivery in the study population [30]. Parturient women means mothers who are in labour or just gave birth. After adjusting for a likely 10% failure of follow-up within the first six months after delivery and allowing for a confidence interval of 2.5 percent with a confidence level of 95 percent, a sample size of 1709 participants was estimated for the study. In essence, the sample size was representative of the Buffalo/Amathole districts in the Eastern Cape Province, South Africa, since the study participants were recruited from three large clinics that cover these two districts. During the study period, all women living with HIV who delivered their babies at the participating hospitals' maternity centers were eligible to take part. There was no refusal from the mothers. The data was collected using a questionnaire that had been pre-validated. It was tested on a group of 20 women who were not part of the main research. The questionnaires were administered by trained research assistants using the face-to-face interview technique. The research assistants had received training on ethical principles in research and interview techniques. Data obtained were captured into an electronic database to facilitate a followup study.

# Variables

### Outcome variable

The outcome variable for this study was a self-reported lifetime induced abortion. Women were asked if they have ever induced abortion and their responses were Yes/ No. All those who responded Yes were coded as "1' and those who responded No were coded as "0".

Variables	All participants	Never induced abortion	Ever induced abortion
All	1709 (100.0)	1384 (81.0)	325 (19.0)
Age			
15–24 years	391 (22.9)	348 (89.0)	43 (11.0)
25 years and over	1318 (77.1)	1036 (78.6)	282 (21.4)
Education level			
Grade 8 or less	120 (7.0)	97 (80.8)	23 (19.2)
Grade 9 to 12	1479 (86.5)	1198 (81.0)	281 (19.0)
Tertiary education	110 (6.4)	89 (80.9)	21 (19.1)
Marital status			
Ever married or cohabiting	522 (30.5)	383 (73.2)	140 (26.8)
Never married	1187 (69.5)	1002 (84.4)	185 (15.6)
Emploment status			
Employed	432 (25.3)	349 (80.8)	83 (19.2)
Unemployed	1277 (74.7)	1035 (81.0)	242 (19.0)
Place of residence			
Rural	585 (34.2)	480 (82.1)	105 (17.9)
Peri-urban	792 (46.3)	639 (80.7)	153 (19.3)
Urban	332 (19.4)	265 (79.8)	67 (20.2)
Parity			
1–2 children	1136 (66.5)	941 (82.8)	195 (17.2)
3–8 children	573 (33.5)	443 (77.3)	130 (22.7)
Smoking status			
Smoked during pregnancy	100 (5.9)	81 (81.0)	19 (19.0)
Quit smoking during pregnancy	80 (4.7)	56 (70.0)	24 (30.0)
Never smoked	1529 (89.5)	1247 (81.6)	282 (18.4)
Alcohol use			
Drank during pregnancy	235 (13.8)	193 (82.1)	42 (17.9)
Stopped drinking during pregnancy	431 (25.2)	343 (79.6)	88 (20.4)
Never drank alcohol	1043 (61.0)	848 (81.3)	195 (18.7)
HIV status at booking			
Positive	1383 (80.9)	1104 (79.8)	279 (20.2)
Negative or unknown	326 (19.1)	280 (85.9)	46 (14.1)

Table 1	Demographic characteristic	s and prevalence of induced ab	ortion among women living with HIV, 2016
---------	----------------------------	--------------------------------	--

### Independent variables

Nine independent variables were considered in the study. The variables were included based on previous studies on factors associated with induced abortion [22, 31]. These comprised ages (15–24, 25 and above), educational level (grade 8 or less, grade 9–12, tertiary education), marital status (Never married, Ever married or cohabiting), employment status (employed, unemployed), place of residence (rural, peri-urban, urban), parity (1–2 children, 3–8 children), smoking status (smoking during pregnancy, quit smoking during pregnancy, never smoked), alcohol use (drank during pregnancy, stopped drinking pregnancy, never drank alcohol) and HIV status at booking (Positive, Negative or unknown).

# Data analyses

Frequencies and percentages were used to describe the participants' sociodemographic characteristics and induced abortion. We fitted two binary logistic regression models. Model I was an unadjusted model that assessed the association between each of the independent variables against the outcome variable (induced abortion). The second model (Model II) was the adjusted model that contained all the independent variables and their association with induced abortion. The Model I and Model II results were presented as unadjusted odds ratios (uOR) and adjusted odds ratios (aOR), respectively, with their corresponding 95% confidence intervals indicating the estimates' level of precision. The study employed binary **Table 2** Unadjusted and adjusted models showing the factorsassociated with induced abortion among women living with HIV,2016

Variables	Model I uOR[95%CI]	Model II aOR[95%CI]
Age		
15-24 years	0.45 (0.32-0.64)***	0.51 (0.35–0.73)***
25 years and over	1	1
Education level		
Grade 8 or less	1.01 (0.52–1.94)	0.98 (0.51–1.93)
Grade 9 to 12	0.99 (0.61–1.63)	0.95 (0.57–1.58)
Tertiary education	1	1
Marital status		
Ever married or cohabiting	1.99 (1.55–2.55)***	1.86 (1.43–2.41)***
Never married	1	1
Employment status		
Employed	1.02 (0.77–1.34)	0.92 (0.69–1.22)
Unemployed	1	1
Place of residence		
Rural	0.87 (0.62–1.22)	0.95 (0.66–1.36)
Peri-urban	0.95 (0.69–1.31)	0.96 (0.69–1.34)
Urban	1	1
Parity		
1–2 children	0.71 (0.55–0.91)*	0.96 (0.73–1.25)
3 to 8 children	1	1
Smoking status		
Ever smoked	1.39 (0.96–2.00)	1.51 (1.01–2.28)*
Never smoked	1	1
Alcohol use		
Ever drank	1.06 (0.82–1.35)	1.02 (0.78–1.35)
Never drank alcohol	1	1
HIV status at booking		
Positive	1.54 (1.10–2.16)*	1.44 (1.02–2.05)*
Negative or unknown	1	1

Exponentiated coefficients; 95% confidence intervals in brackets; \*p < 0.05, \*\*p < 0.01, \*\*\*p < 0.001; 1 = reference

logistic regression analysis because the outcome variable was dichotomous. In all the analyses, statistical significance was set at a p-value less than 0.05. Variance inflation factor was used to assess multicollinearity among the variables. All the analyses were carried out by using the IBM SPSS Statistics for Windows, Version 27.0 (IBM Corp., Armonk, New York, USA).

#### **Ethical consideration**

We followed various steps to ensure the study adhered to all the ethical principles in line with the Declaration of Helsinki. First, the Walter Sisulu University Ethics Committee (reference number 098/2014) and the Eastern Cape Department of Health gave their approval upon assessing the research protocol before the data collection commenced. Second, the study procedures were approved by the management of all the three hospitals. Furthermore, the study's purpose and process were explained to all the participants in an information sheet written in English and translated into IsiXhosa. Each participant gave written, informed consent to indicate her voluntary participation in the study. We obtained parental consent for three participants below 16 years, who assented to participants to privacy, confidentiality, and anonymity throughout the study.

# Results

# **Descriptive findings**

The average age of participants was  $29.63\pm6.17$  years. Most participants were over 25 years (77.1%), had grade 9 to 12 level of education (86.5%), unemployed (74.7%), and never married (69.5%). The prevalence of induced abortion was 19% (95% CI: 17.2–20.9%) but varied by sociodemographic characteristics. Induced abortion prevalence was higher among women aged 25 years and over (21.4%), ever married (26.8%), those already diagnosed positive before their index pregnancy than among women aged less than 25 years, never-married women, and those diagnosed with HIV during their index pregnancy.

#### **Multivariable findings**

In the unadjusted logistic regression model, age 15-24 years, ever married, parity 1-2, and being diagnosed with HIV before the index pregnancy were significantly associated with having ever (lifetime) induced abortion. Education, employment, place of residence, and alcohol use were not significantly related to ever having an induced abortion. In the adjusted logistic regression model, all variables that were significant in the unadjusted regression maintained their direction and magnitude of effects, except for parity. Adolescents and young adults (aOR:0.51; 95% CI:0.35-0.73) had lower odds of having ever induced abortion, compared with women aged 25 years and above. Ever married or cohabiting women(aOR:1.86; 95% CI:1.43-2.41) had higher odds of having ever induced abortion compared with nevermarried women. Women who had one or two children were less likely to have induced abortion in comparison to those with three or more children. Women who ever smoked (aOR: 1.51; 95% CI: 1.01-2.28) were more likely to have induced abortion compared with those who never smoked. Women diagonosed positive with HIV before their index pregnancy (aOR:1.44; 95% CI: 1.02-2.05) were more likely to have induced abortion relative to those diagnosed during their index pregnancy.

# Discussion

The present study sought to assess the prevalence and factors associated with induced abortion among women living with HIV from three large health facilities in South Africa. This is the first study to examine the prevalence and factors associated with induced abortion among women living with HIV in South Africa to the best of our knowledge. We found that the prevalence of induced abortion was 19% but varied by sociodemographic characteristics. Our study's prevalence of induced abortion suggests that the permissive legal context of abortion in South Africa may have helped some women access abortion upon demand [25, 32]. The findings from this study are vital in the sense that about 19 out of every 100 women who participated in this study have ever had an induced abortion. This reiterates the point that a greater proportion of these women need comprehensive abortion care services. It also draws attention to the fact that most of these women are experiencing unintended pregnancies, and hence their need for abortion care. Therefore, the more we report on the number of women living with HIV having abortions, the more healthcare providers will begin to understand that these women need abortion-related care. As such, results on abortion among this core group can help plan the various steps necessary for abortion care for the teeming population of women who need the service in the country.

The prevalence of induced abortion reported in the current study is lower than what was reported in previous studies in Brazil (30%) [33] and Italy (41.4%) [34]. It is worth noting that data of these studies were collected before the existence of highly active antiretroviral therapy (HAART) and when the vertical transmission prevention protocols were still starting, which restricts the comparability of the results to our study. Prevalence of induced abortion would expectedly drop significantly after the availability of HAART. Darak, et al. [31] studied 158 pregnancies of women after their HIV diagnoses and found that 51% were unwanted and 50% were induced abortion. The prevalence of induced abortion in our study is higher than what Pilecco et al. [22] and Pinho et al. [23], found (6.5% and 11.9%, respectively) among women living with HIV in Brazil. But these studies evaluated all pregnancies in contrast to our study that evaluated lifetime prevalence of abortion.

Another significant finding of this study is that adolescents and young adults were 49% less likely to have ever induced abortion compared with women aged 25 and above. This finding is consistent with the literature [35, 36]. This result is expected given that we measured the lifetime prevalence of induced abortion. Older people have more years of exposure to unintended pregnancies compared to younger ones due to their duration of sexual experience.

We also observed no socioeconomic variations (education, residence and employment status) in terms of the odds of pregnancy termination. This result contrasts with several previous studies in other parts of Africa that have reported socioeconomic and geographical inequalities in access to reproductive health services [35, 37]. The possible explanation for our finding could be the free healthcare policy in South Africa and availability of abortion services at the district hospitals across the region that has taken off the financial burden for the poor and therefore eliminated disparities between and within various groups in terms of healthcare access [38, 39].

Our study also shows that married/cohabiting women had higher odds of having ever induced abortion compared with never-married women. This finding contrasts with a previous study among women without HIV that reported a higher prevalence among single women [36]. It is worth indicating that abortion is not the risk itself; however, it is the unintended pregnancy. Evidence also shows a high prevalence of unintended pregnancy among women living with HIV in South Africa [14-16]. Married and cohabiting women are perhaps more sexually active than single women. Their exposure to sex means that their risk of unintended pregnancy is higher than single women, especially if they are not using effective contraceptives consistently [31]. Another possible reason is that married and cohabiting women may be older than the unmarried ones and might already achieve their fertility desire and therefore not want another child. If a woman become pregnant when she had achieved her fertility desire, she would most likely terminate the pregnancy, unlike if she has not reached her fertility goal.

Although not statistically significant in the adjusted model, women with parity of one or two had lower odds of having ever induced abortion than those with three to eight children. A previous study among women with HIV reported a similar result [31]. This finding is not surprising because the fertility rate per couple is 2.4 children [40] in South Africa. Those with fewer children might want to have more compared to those who had fulfilled their fertility desire. Women who have reached their ideal family size may choose to terminate an unwanted pregnancy to restrict their family size [41].

Lastly, our study shows that women diagnosed with HIV before their index pregnancy had higher likelihood of having ever induced abortion compared to those diagnosed during their index pregnancy. The literature has reported mixed results on whether the prevalence of induced abortion is higher among women with or without HIV. While a study that analysed data from Nigeria and Zambia shows that the odds of induced abortion did not differ among women with and without HIV [42], evidence from a recent systematic review [10] and surveys [20, 23] reported higher prevalence among women living with HIV than those without HIV. Fear of transmitting the virus to the baby [20, 21], exposing the child to later stigma, and discrimination, and leaving a child orphaned [20, 21] may have contributed to the relatively higher prevalence of induced abortion among women living with HIV.

#### **Strength and limitations**

The major limitation of the present study is that abortion is self-reported. Due to abortion related stigma, some women may have induced abortion but fail to report it. Therefore, the prevalence of induced abortion in our study is conservative. Also, our study design was crosssectional, and due to this, it is highly impossible to establish causal relationships. Because our study relied on past abortion experiences relative to current socioeconomic, our findings on the association between socioeconomic characteristics and abortion should be interpreted with caution. It is possible that sub-group of women needing more interventions are not identified due to the limitation of our dataset. Despite these limitations, the strengths are also worth acknowledging. First, the sample size is relatively large, and this has given us the statistical power to estimate factors associated with induced abortion among women with HIV. Again, the study was conducted in three large hospitals with diverse respondents from different backgrounds.

#### Conclusion

The prevalence of induced abortion among parturient women living with HIV who participated in this study is relatively high. This is an indication that the need for abortion is somewhat high in this group of women. The finding, therefore, highlights the need for concerted efforts from all stakeholders to address the unmet need for contraception among women living with HIV to prevent unintended pregnancy. It is pertinent to address all barriers hindering access to comprehensive abortion care to ensure women use approved health facilities for this important reproductive health service.

#### Abbreviations

cOR: Crude odds ratios; aOR: Adjusted odds ratio; CI: Confidence interval.

#### Acknowledgements

We acknowledge the study participants for their participation in the study. We also appreciate the research assistants—Ulama Sidloyi, Shulie Majambe, Elizabeth Bosha, Sandi Mzilikazi, Prince Sofute, and Sandile Mqwashini—for their contribution towards a successful data collection.

#### Author contributions

AIA and OVA conceptualised the study. AIA performed the analysis. BOA, AIA, AS and OVA contributed to the draft and approved the final version. All authors read and approved the final manuscript.

#### Funding

This work was supported by the South African AIDS Vaccine Initiative under the auspices of the South African Medical Research Council [Grant Number: SAMRC/03/2017] with funds received from the South African National Department of Health. OVA is supported by research grants from the Discovery Foundation, South Africa [Grant Number: 035968/2016]. AIA's time to write this article was supported through the funding to the African Population and Health Research Center from the Swedish International Development Cooperation Agency for the Challenging the Politics of Social Exclusion project (Sida Contribution No. 12103). Paediatric seeds support was received from the University College Dublin. Funders have no direct involvement in the implementation of the project or content of the manuscripts.

#### Data availability

Data will be made available by the corresponding author upon reasonable request.

#### Declarations

#### Ethics approval and consent to participate

We followed various steps to ensure the study adhered to all the ethical principles in line with the Declaration of Helsinki. First, the Walter Sisulu University Ethics Committee (reference number 098/2014) and the Eastern Cape Department of Health gave their approval upon assessing the research protocol before the data collection commenced. Second, the study procedures were approved by the management of all the three hospitals. Furthermore, the study's purpose and process were explained to all the participants in an information sheet written in English and translated into IsiXhosa. Each participant gave written, informed consent to indicate her voluntary participation in the study. We obtained parental consent for three participants below 16 years, who assented to participate in the study. We respected the rights of participants to privacy, confidentiality, and anonymity throughout the study.

#### **Consent for publication**

No consent to publish was needed for this study as we did not use any details, images or videos related to individual participants.

#### **Competing interests**

The authors declare no competing interests.

#### Author details

<sup>1</sup> Sexual, Reproductive, Maternal, New-born, Child and Adolescent Health (SRMNCAH) Unit, African Population and Health Research Center, APHRC Campus, Manga Close, Nairobi, Kenya. <sup>2</sup>School of Public Health, Faculty of Health, University of Technology, Sydney, Australia. <sup>3</sup>Centre for Gender and Advocacy, Takoradi Technical University, Takoradi, Ghana. <sup>4</sup>College of Public Health, Medical and Veterinary Sciences, James Cook University, Townsville, QLD, Australia. <sup>5</sup>Department of Family Medicine, East London Hospital Complex, Walter Sisulu University/Cecilia Makiwane Hospital, East London, South Africa.

# Received: 27 June 2021 Accepted: 1 November 2022 Published online: 05 December 2022

#### References

- 1. UNAIDS. Seizing the moment: tackling entrenched inequalities to end epidemics. Global, AIDS update 2020. Geneva: Joint United Nations Programme on HIV/AIDS (UNAIDS); 2020.
- Juliastuti D, Dean J, Fitzgerald L. Sexual and reproductive health of women living with HIV in Muslim-majority countries: a systematic mixed studies review. BMC Int Health Hum Rights. 2020;20(1):1–12.
- Teixeira LB, Pilecco FB, Vigo A, et al. Sexual and reproductive health of women living with HIV in Southern Brazil. Cad Saude Publica. 2013;29(3):609–20.

- 4. Erhabor O, Adias T, Akani C. Reproductive health challenges of living with HIV-infection in Sub Saharan Africa. Curr Perspect HIV Infect 2013:325.
- Gutin SA, Namusoke F, Shade SB, et al. Fertility desires and intentions among HIV-positive women during the post-natal period in Uganda. Afr J Reprod Health. 2014;18(3):67–77.
- Wekesa E, Coast E. Fertility desires among men and women living with HIV/AIDS in Nairobi slums: a mixed methods study. PLoS ONE. 2014;9(8): e106292.
- Mumah JN, Ziraba AK, Sidze EM. Effect of HIV status on fertility intention and contraceptive use among women in nine sub-Saharan African countries: evidence from Demographic and Health Surveys. Glob Health Action. 2014;7(1):25579.
- UNAIDS. We can prevent mothers from dying and babies from becoming infected with HIV. Joint Action for Results UNAIDS Outcome Framework: Business Case 2009–2011. Geneva: UNAIDS; 2010.
- Adeniyi OV, Ajayi AI, Selanto-Chairman N, et al. Demographic, clinical and behavioural determinants of HIV serostatus non-disclosure to sex partners among HIV-infected pregnant women in the Eastern Cape, South Africa. PLoS ONE. 2017;12(8): e0181730.
- Feyissa TR, Harris ML, Melka AS, et al. Unintended pregnancy in women living with HIV in Sub-Saharan Africa: a systematic review and metaanalysis. AIDS Behav. 2019;23(6):1431–51.
- Kikuchi K, Wakasugi N, Poudel KC, et al. High rate of unintended pregnancies after knowing of HIV infection among HIV positive women under antiretroviral treatment in Kigali, Rwanda. Biosci Trends. 2011;5(6):255–63.
- Mayondi GK, Wirth K, Morroni C, et al. Unintended pregnancy, contraceptive use, and childbearing desires among HIV-infected and HIV-uninfected women in Botswana: across-sectional study. BMC Public Health. 2015;16(1):1–10.
- McCoy SI, Buzdugan R, Ralph LJ, et al. Unmet need for family planning, contraceptive failure, and unintended pregnancy among HIV-infected and HIV-uninfected women in Zimbabwe. PLoS ONE. 2014;9(8): e105320.
- Credé S, Hoke T, Constant D, et al. Factors impacting knowledge and use of long acting and permanent contraceptive methods by postpartum HIV positive and negative women in Cape Town, South Africa: a crosssectional study. BMC Public Health. 2012;12(1):1–9.
- Schwartz SR, Rees H, Mehta S, et al. High incidence of unplanned pregnancy after antiretroviral therapy initiation: findings from a prospective cohort study in South Africa. PLoS ONE. 2012;7(4): e36039.
- Adeniyi OV, Ajayi AI, Moyaki MG, et al. High rate of unplanned pregnancy in the context of integrated family planning and HIV care services in South Africa. BMC Health Serv Res. 2018;18(1):140.
- Iyun V, Brittain K, Phillips TK, et al. Prevalence and determinants of unplanned pregnancy in HIV-positive and HIV-negative pregnant women in Cape Town, South Africa: a cross-sectional study. BMJ Open. 2018;8(4): e019979.
- Kaida A, Laher F, Strathdee SA, et al. Contraceptive use and method preference among women in Soweto, South Africa: the influence of expanding access to HIV care and treatment services. PLoS ONE. 2010;5(11): e13868.
- WHO. Safe abortion care: the public health and human rights rationale. Safe Abortion 2012;19. Retrieved from 5 May 2021. https://repro ductiverights.org/who-report-safe-abortion-is-essential-to-prote cting-womens-human-rights/
- Chi BK, Gammeltoft T, Nam NTT, et al. Induced abortion among HIVpositive women in Quang Ninh and Hai Phong, Vietnam. Trop Med Int Health. 2010;15(10):1172–8.
- Orner P, de Bruyn M, Cooper D. 'It hurts, but I don't have a choice, I'm not working and I'm sick': decisions and experiences regarding abortion of women living with HIV in Cape Town, South Africa. Cult Health Sex. 2011;13(7):781–95.
- De Bruyn M. Living with HIV: challenges in reproductive health care in South Africa. Afr J Reprod Health. 2004;8(1):92–8.
- Pilecco FB, Teixeira LB, Vigo A, et al. Lifetime induced abortion: a comparison between women living and not living with HIV. PLoS ONE. 2014;9(4): e95570.
- Pinho ADA, Villela WV, Barbosa RM, et al. Abortion among women living with or not living with HIV/AIDS users of public health services in São Paulo municipality: prevalence, contexts and reasons. Rev Bras Saude Materno Infant. 2019;19(4):837–49.

- 25. Albertyn C. Claiming and defending abortion rights in South Africa. Revista Direito GV. 2015;11(2):429–54.
- Harries J, Cooper D, Strebel A, et al. Conscientious objection and its impact on abortion service provision in South Africa: a qualitative study. Reprod Health. 2014;11(1):1–729.
- 27. Suryavanshi N, Erande A, Pisal H, et al. Repeated pregnancy among women with known HIV status in Pune, India. AIDS Care. 2008;20(9):1111–8.
- South African National Aids Council. Women's sector report on UNGASS HIV and AIDS declaration goals: UNGASS Forum: report on progress towards the UNGASS goals related to sexual and reproductive health and rights. Cape Town: Mosaic: South Africa National Aids Council, 2010.
- Africa SS. Buffalo city metropolitan municipality, 2011.Retrived from 5 May 2021. https://www.google.com/url?sa=t&rct=j&q=&esrc=s& source=web&cd=&cad=rja&uact=8&ved=2ahUKEwi\_-ljfk6f7AhVl2DgG HWBjAQkQFnoECA8QAQ&url=https%3A%2F%2Fwww.statssa.gov.za% 2F%3Fpage\_id%3D1021%26id%3Dbuffalo-city-municipality&usg= AOvVaw0yT6cydguwAHjrB6w7EGtT
- Bedelu M, Ford N, Hilderbrand K, et al. Implementing antiretroviral therapy in rural communities: the Lusikisiki model of decentralized HIV/ AIDS care. J Infect Dis. 2007;196(Supplement 3):S464–8.
- Darak S, Hutter I, Kulkarni V, et al. High prevalence of unwanted pregnancies and induced abortions among HIV-infected women from Western India: Need to emphasize dual method use? AIDS Care. 2016;28(1):43–51.
- BerroPizzarossa L, Durojaye E. International human rights norms and the South African choice on termination of pregnancy act: an argument for vigilance and modernisation. S Afr J Human Rights. 2019;35(1):50–69.
- Domingues RMSM, Silva CMFPD, Grinsztejn BGJ, et al. Prevalence of induced abortion and associated factors in a cohort of women living with HIV/AIDS, Rio de Janeiro, Brazil, 1996–2016. Cad Saude Publica. 2020;36:e00201318.
- Ammassari A, Cicconi P, Ladisa N, et al. Induced first abortion rates before and after HIV diagnosis: results of an Italian self-administered questionnaire survey carried out in 585 women living with HIV. HIV Med. 2013;14(1):31–9.
- 35. Alemayehu B, Addissie A, Ayele W, et al. Magnitude and associated factors of repeat induced abortion among reproductive age group women who seeks abortion Care Services at Marie Stopes International Ethiopia Clinics in Addis Ababa, Ethiopia. Reprod Health. 2019;16(1):1–10.
- Baruwa OJ, Amoateng AY, Biney E. Induced abortion in Ghana: prevalence and associated factors. J Biosoc Sci. 2021. https://doi.org/10.1017/S0021 932020000772.
- Tesema GA, Mekonnen TH, Teshale AB. Spatial distribution and determinants of abortion among reproductive age women in Ethiopia, evidence from Ethiopian Demographic and Health Survey 2016 data: Spatial and mixed-effect analysis. PLoS ONE. 2020;15(6): e0235382.
- Barron P, Padarath A. Twenty years of the South African health review. S Afr Health Rev. 2017;2017(1):1–10.
- Hlafa B, Sibanda K, Hompashe DM. The impact of public health expenditure on health outcomes in South Africa. Int J Environ Res Public Health. 2019;16(16):2993.
- 40. Health NDo, Africa SS, Council SAMR, et al. South Africa Demographic and Health Survey 2016: NDoH, Stats SA, SAMRC, and ICF Pretoria, South Africa, and Rockville ..., 2019.
- Tesfaye G, Hambisa MT, Semahegn A. Induced abortion and associated factors in health facilities of Guraghe zone, southern Ethiopia. J Pregnancy. 2014. https://doi.org/10.1155/2014/295732.
- Bankole A, Keogh S, Akinyemi O, et al. Differences in unintended pregnancy, contraceptive use and abortion by HIV status among women in Nigeria and Zambia. Int Perspect Sex Reprod Health. 2014;40(1):28–38.

#### **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.